Author’s response to reviews

Title: DiAlert: a lifestyle education programme aimed at people with a positive family history of type 2 diabetes and overweight, study protocol of a randomised controlled trial.

Authors:

Wieke H Heideman (w.heideman@vumc.nl)
Vera Nierkens (v.nierkens@amc.uva.nl)
Karien Stronks (k.stronks@amc.uva.nl)
Barend JC. Middelkoop (b.middelkoop@lumc.nl)
Jos WR Twisk (jwr.twisk@vumc.nl)
Arnoud P Verhoeff (averhoeff@ggd.amsterdam.nl)
Maartje de Wit (m.dewit@vumc.nl)
Frank J Snoek (fj.snoek@vumc.nl)

Version: 3 Date: 21 September 2011

Author’s response to reviews: see over
Dear Natalie Pafitis MSc,

Please find enclosed the revised manuscript entitled ‘DiAlert: a lifestyle education programme aimed at people with a positive family history of type 2 diabetes and overweight, study protocol of a randomised controlled trial’.

In response to your comment: Document, within your manuscript, whether you will receive informed consent from your study participants and whether this will be verbal or written.

We supplemented the manuscript with this information on page 6: Stating: After receiving a signed informed consent form from the participants, randomisation will be performed with sealed envelopes.

In addition, we revised the sentence about ethical approval on page 11 stating: The study protocol, information letters and informed consent form were approved by the Medical Ethics Review Committee of the VU University Medical Center.

Conform the journal style we included Table 1 at the end of the manuscript.

We hope that after our revision, you will find the manuscript of sufficient quality to merit publication in BMC Public Health.

Sincerely yours,

W.H. Heideman, MSc.

Corresponding author:

Ms. Wieke Heideman, MSc.
VU University Medical Centre, Department of Medical Psychology, Diabetes Psychology Research Group
Van der Boechorststraat 7, MF G-417, 1081 BT Amsterdam, The Netherlands
Tel: +31 (0)20 4445682
Fax: +31 (0)20 4448230
E-mail: w.heideman@vumc.nl